Author’s response to reviews

Title: A real-world study of effectiveness of intravitreal bevacizumab and ranibizumab injection for treating retinal diseases in Thailand

Authors:

Suthasinee Kumluang (suthasinee.k@hitap.net)
Lily Ingsrisawang (fscilli.ing@gmail.com; fscilli@ku.ac.th)
Sermsiri Sangroongruangsri (sng_ning@hotmail.com)
Usa Chaikledkaew (usa.chi@mahidol.ac.th)
Tanapat Ratanapakorn (tratanapakorn@gmail.com)
Paisan Ruamviboonsuk (paisan.trs@gmail.com)
Wongsiri Taweebanjongsin (wongsiri156@gmail.com)
Janejit Choovuthayakorn (janejit.c@cmu.ac.th)
Somanus Thoongsuwan (somanus@hotmail.com)
Prut Hanutsaha (prut.han@mahidol.ac.th)
Kittisak Kulvichit (kulvichit@gmail.com)
Thitiporn Ratanapojnard (tratanaj@hotmail.com)
Warapat Wongsawad (warroph@gmail.com)
Pattara Leelahavarong (pattara.l@hitap.net)
Yot Teerawattananon (yot.t@hitap.net)

Version: 4 Date: 28 Feb 2019

Author’s response to reviews:

Dear Editors,

On behalf of my co-authors, I would like to resubmit a revised version of manuscript entitled “A real-world study of effectiveness of intravitreal bevacizumab and ranibizumab
injection for treating retinal diseases in Thailand” (BOPH-D-18-00239R4) for publication in “BMC Ophthalmology”.

We would like to express our sincere gratitude to the editors for giving us the opportunity to the revise and resubmit our manuscript. We also appreciate your precise comments and we have made corrections our manuscript according to the comments.

Please find enclosed the revised version. We hope that the revised manuscript will meet your requirements and be acceptable for publication. Thank you very much once again for your kind consideration of our revised manuscript. Should you have any further requests or questions, please do not hesitate to contact me.

We look forward to hearing from you soon.

Sincerely yours,

Assist. Prof. Lily Ingsrisawang, Ph.D.
Corresponding author
Department of Statistics, Faculty of Science, Kasetsart University, Bangkok, 10900, Thailand
Tel: 662-5625444, 662-5625555 ext 647101
Fax: 662-9428384
Email: fscilli@ku.ac.th, fscilli.ing@gmail.com